Author's response to reviews

Title: Exome sequencing reveals a novel TTC19 mutation in an autosomal recessive spinocerebellar ataxia patient

Authors:

Hiroyuki Morino (morino@hiroshima-u.ac.jp)
Ryosuke Miyamoto (miyamoto@clin.med.tokushima-u.ac.jp)
Shizuo Ohnishi (ohnishi@tanabe-neuro-hsp.jp)
Hirofumi Maruyama (hmaru@hiroshima-u.ac.jp)
Hideshi Kawakami (hkawakam@hiroshima-u.ac.jp)

Version: 5 Date: 20 December 2013

Author's response to reviews: see over
December 20, 2013
Mr. Josefino Rodis
Editor, BMC Neurology
Floor 6, 236 Gray's Inn Road
London, WC1X 8HB, UK

MS ID: 1804922176107544

Dear Mr. Rodis;

Thank you for your email of November 21, 2013, regarding our manuscript, "Exome sequencing reveals a novel TTC19 mutation in an autosomal recessive spinocerebellar ataxia patient," as well as the favorable comment of the reviewer.

We sent an email about the second review on December 2. We are concerned whether or not you could receive it, so we send a reply from the BioMed Central website again. We would appreciate if you could confirm it.

Regarding the language corrections, we have had Nature Publishing Group language editing service correct our manuscript twice, once before the initial submission, and the next before the last revision. Attached please find the certifications of the proofreading. If you do not mind, we would like you to indicate the specific passage of the manuscript we should improve. Thank you in advance for your kind consideration of this paper.

All very best wishes,

Hiroyuki Morino, MD, PhD
Department of Epidemiology, RIRBM
Hiroshima University
1-2-3 Kasumi Minami-ku, Hiroshima, 734-8553, Japan

E-mail: morino@hiroshima-u.ac.jp
TEL: +81-82-257-5849, FAX: +81-82-257-5848